Author's response to reviews

Title: Increased Risk of Affective Disorders in Type 2 Diabetes Is Minimised by Sulfonylurea and Metformin Combination: A Population-Based Cohort Study

Authors:

Mark L. Wahlqvist (profmlw@nhri.org.tw)
Meei-Shyuan Lee (mmsl@ndmctsgh.edu.tw)
Shao-Yuan Chuang (chuangsy@nhri.org.tw)
Chih-Cheng Hsu (cch@nhri.org.tw)
Hsin-Ni Tsai (970721@nhri.org.tw)
Shu-Han Yu (shuhan229@gmail.com)
Hsing-Yi Chang (hsingyi@nhri.org.tw)

Version: 9 Date: 2 November 2012

Author's response to reviews:

November 2, 2012
Ms. Ref. No.: 1223292856732899

Dear Editor,

According to your requests, we have made the following changes.

1. Please ensure that the title is in sentence case.
Ans: It has been changed.

2. Please ensure that each author affiliation contains, in the following order, department, institution, street name and number, town, post code and country.
Ans: Done.

3. Please remove all issue numbers from the reference list.
Ans: Done. We have also updated all references with publication details.

4. Please note that each reference can contain up to 30 authors' names for each citation. The term "et al." should not be used if there are fewer than 30 authors.
3. Please move the tables into the manuscript file, after the figure legends. The tables should then be removed as additional files from the submission system.
Ans: Done.

4. Tables must be created using a Tables tool, and should not be linked files, figures, or generated with tabbed text. Tables must be divided into rows and columns and be oriented in a portrait page layout. Please ensure that text alignment in columns is clear and consistent. We recommend the MS Word auto-format template: simple 1. Please do not include vertical lines in your tables.
Ans: We have following the required rules.

5. Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance.
Ans: Done.

With kind regards,

Mark L Wahlqvist MD FRACP